Dear Editor,

Thank you for your consideration of our manuscript. Please find attached a revised version.

This version was checked out by a native English speaker (specialist in linguistic and communication), the language was revised, and modifications made to overcome the requested high standards.

Age, gender and ethnicity were added to the abstract section.

Other modifications were made to conform to the journal style.

Hope these revisions respond to your requests.

Sincerely,

Mohamed Mesmoudi.

Department of medical oncology, National institute of oncology, Rabat, Morocco.

mohamedmesmoudi@yahoo.fr